

Efficient Bayesian Inference for Multiple Change-Point and Mixture Innovation Models

Paolo Giordani and Robert Kohn

MAY 2006

WORKING PAPERS ARE OBTAINABLE FROM

Sveriges Riksbank • Information Riksbank • SE-103 37 Stockholm Fax international: +46 8 787 05 26 Telephone international: +46 8 787 01 00 E-mail: info@riksbank.se

The Working Paper series presents reports on matters in the sphere of activities of the Riksbank that are considered to be of interest to a wider public. The papers are to be regarded as reports on ongoing studies and the authors will be pleased to receive comments.

The views expressed in Working Papers are solely the responsibility of the authors and should not to be interpreted as reflecting the views of the Executive Board of Sveriges Riksbank.

Efficient Bayesian Inference for Multiple Change-Point and Mixture Innovation Models^{*}

Paolo Giordani[†] Research Department Sveriges Riksbank Robert Kohn School of Economics School of Banking and Finance University of New South Wales

Sveriges Riksbank Working Paper Series No. 196 May 2006

Abstract

Time series subject to parameter shifts of random magnitude and timing are commonly modeled with a change-point approach using Chib's (1998) algorithm to draw the break dates. We outline some advantages of an alternative approach in which breaks come through mixture distributions in state innovations, and for which the sampler of Gerlach, Carter and Kohn (2000) allows reliable and efficient inference. We show how this approach can be used to (i) model shifts in variance that occur independently of shifts in other parameters (ii) draw the break dates efficiently in change-point and regime-switching models with either Markov or non-Markov transition probabilities. We extend the proofs given in Carter and Kohn (1994) and in Gerlach, Carter and Kohn (2000) to state-space models with system matrices which are functions of lags of the dependent variables, and we further improve the algorithms in Gerlach, Carter and Kohn by introducing to the time series literature the concept of adaptive Metropolis-Hastings sampling for discrete latent variable models. We develop an easily implemented adative algorithm that promises to sizably reduce computing time in a variety of problems including mixture innovation, change-point, regime-switching, and outlier detection.

Keywords: Structural breaks; Parameter instability; Change-point; State-space; Mixtures; Discrete latent variables; Adaptive Metropolis-Hastings.

JEL Classification Codes: C11, C15, C22.

^{*}We thank John Haywood and Gary Koop for comments. The views expressed in this paper are those of the authors and should not be interpreted as reflecting the views of the Executive board of Sveriges Riksbank.

[†]Corresponding author: paolo.giordani@riksbank.se

1 Introduction

This paper is concerned with the problem of modeling and inference for processes subject to random shifts in parameters at unknown dates. The literature on the topic can be traced back at least to Quandt (1958) and Chernoff and Zack (1964), but has expanded rapidly in recent years because of faster computers and the availability of powerful new statistical tools for the solution of the considerable numerical problems involved. Additional incentives have been provided by Clements and Hendry (1999), who point to failures to detect (or to properly adjust to) intercept shifts as a main source of forecast failure, and by Stock and Watson (1996), who document the widespread nature of parameter instability in linear models of U.S. macroeconomic and financial time series.

Traditionally, a majority of papers in this area take a frequentist approach.¹ However, in recent years the number of Bayesian alternatives has increased rapidly, as many authors came to recognize the theoretical and practical advantages of a Bayesian approach for these types of problems. From a theoretical perspective, it seems desirable for parameter and forecast distributions to reflect uncertainty on the number and timing of break-points, rather than being conditioned on modal values. From a practical stance, because the likelihood quickly becomes intractable as the number of breaks increase², frequentists are often forced to resort to simulation methods similar to those employed by Bayesians, therefore losing the advantage of simpler implementation enjoyed in other models.

The Bayesian approach coupled with modern simulation methods makes it tractable to estimate models with multiple structural breaks. Latent variables determine the location and nature of the breaks. Conditional on these latent variables, it is possible to generate the model parameters, while conditional on the parameters it is possible to generate the latent variables. Such generation is repeated until an adequate sample is obtained from the posterior distribution of interest. The most popular Bayesian approach in the econometric literature on structural breaks follows Chib (1998), who models the breaking process as a Markov chain with transition probabilities constrained so that regimes come in a non-reversible sequence. (We refer to this as a 'change-point' approach.) Pesaran, Pettenuzzo, and Timmermann (2004) extend the results in Chib (1998) by adding a hierarchical level for all parameters. Koop and Potter (2004a, 2004b) make a significant advance by allowing for an unknown number of breaks.

 $^{^1 \, {\}rm The}$ frequentist literature on testing for structural breaks is reviewed by Hansen (2001) and Elliott and Müller(2003).

²See the discussion in Elliott and Müller (2003).

An alternative approach builds on the state-space representation, modeling the breaking process through mixture distributions for the state innovations. (We refer to this as a 'mixture innovation' approach.) Time-varying-parameter models with normal innovations in state variables are well-known and easily estimated special cases (see Cogley and Sargent (2001) and Primiceri (2005) for recent Bayesian economic applications), but the general case of several component distributions presents formidable computational problems. Several approximations have been proposed, including Harrison and Stevens (1976), Sims (1993) and Engle and Smith (1999). Markov chain Monte Carlo techniques opened the way for the exact solutions developed by McCulloch and Tsay (1993), Carter and Kohn (1994), and Shephard (1994). However, these algorithms are at best extremely inefficient in the case of relatively infrequent breaks. The reason is that they draw the discrete latent variables conditional on the states, and in models with structural breaks the correlation between these two blocks is very high or even perfect. For models that can be written in conditionally Gaussian state-space form, the sampling algorithm of Gerlach, Carter and Kohn (2000) avoids this problem and permits fast and efficient estimation of mixture innovation models.

Our paper makes five contributions to the Bayesian literature on structural breaks. The first is to argue that mixture innovations in state variables are often an intuitive and flexible way of modelling breaks, and that even complex models can be estimated efficiently with the method of Gerlach, Carter and Kohn (2000). Mixture innovation models have a number of desirable features, including a random number of in-sample breaks, the possibility of jointly modeling small (and more frequent) and large (and less frequent) breaks, easy inclusion of innovation and additive outliers, ease of prior elicitation, and the relative convenience of allowing different parameters to change at different times.

The second contribution is to show how to perform efficient inference for models with breaks in conditional variance, and how to allow breaks in the parameters of the conditional mean and conditional variance to occur independently.

Of course in some instances a change-point approach may still be preferred. Koop and Potter (2004b) is then an interesting option because it allows for a random number of breaks. However, it requires $O(n^2)$ operations to draw the break dates with a geometric prior distribution for regime durations, and $O(n^3)$ operations with a Poisson prior (their preferred choice). The third contribution of our paper is to show how to draw the break-points in the model of Koop and Potter (2004b) in just O(n) operations with either prior. More generally, we show how to perform efficient Bayesian inference for state-space models with latent variables having non-Markov transition probabilities.

The fourth contribution is to introduce to the time series literature the concept of adaptive Metropolis-Hastings sampling for discrete latent variable models. We present a simple algorithm which further improves sampling efficiency (over Gerlach *et al.*, 2000) in a class of discrete latent variable models that includes change-point, mixture innovation, regime switching, and outlier detection. The adaptive algorithm uses past draws of the discrete latent variables to design a proposal distribution for a Metropolis-Hastings step, greatly reducing computing time spent (and mostly wasted) on observations for which the presence or absence of a break is rather clear-cut. In two applications to U.S. inflation and real interest rate, this adaptive algorithm reduces computing times (compared to Algorithm 1 in Gerlach *et al.* (2000), for a given Monte Carlo standard error) by over 70% and 60% respectively.

The proofs in Carter and Kohn (1994) and in Gerlach, Carter and Kohn (2000) do not include models in which the system matrices are functions of lags of the dependent variables. However, as it turns out, both algorithms still apply. The fifth contribution of this paper is to show that the proofs generalize to this class of models. The proofs in Gerlach, Carter and Kohn (2000) are also extended to the multivariate case.

The paper is organized as follows. Section 2 presents three Bayesian approaches to structural change, and discusses some advantages of mixture innovation models. In particular, it argues that the method of Gerlach *et al.* (2000) solves the computational problems that have so far constrained their use and development. Section 3 suggests a method to model breaks in variances in conditionally Gaussian form and to combine shifts in conditional mean and variance parameters. Section 4 shows how the sampler of Gerlach *et al.* (2000) can be adapted to draw the break dates in models with non-Markov breaking processes, using the case of a Poisson prior for regime duration as an example. Section 5 outlines a new adaptive Metropolis-Hastings sampler which promises to further increase sampling efficiency in a variety of discrete latent variable problems. Section 6 evaluates the efficiency gains from the new adaptive sampler on models of U.S. inflation and real interest rate. Section 7 concludes the paper. An appendix generalizes the proofs of Carter and Kohn (1994) and Gerlach, Carter and Kohn (2000).

2 Bayesian approaches to structural change

This section reviews three approaches to modeling structural change: changepoint models, time-varying-parameter models, and mixture innovation models. For our purposes, it is useful to cast all these models in a common framework, writing them in state-space form, with system matrices depending on vectors of discrete latent variables. The contributions of our article are for conditionally Gaussian models. Conditionally Gaussian models can be written in linear and Gaussian state-space form conditional on a vector of latent variables³

$$y_t = g_t + h'_t x_t + \gamma_t u_t \tag{1}$$

$$x_t = f_t + F_t x_{t-1} + \Gamma_t v_t, \tag{2}$$

where $u_t \sim nid(0, 1)$, $v_t \sim nid(0, I)^4$ and all system matrices at time t are known conditional on parameters, on lagged values of y_t , on exogenous covariates, and on a vector of discrete latent variables K_t . Throughout the paper, for any variable z_t , we define $z^{t_1,t_2} = (z_{t_1},...,z_{t_2})$ and $z = (z_1,...,z_n)$. The latent variables are typically assumed to be independent or Markov. Most time series models take a conditionally Gaussian state-space form, and need not be Gaussian after integrating out the latent variables. In fact, besides models for structural change, the most common nonlinear models and several models for fat-tailed errors are conditionally Gaussian (see Giordani, Kohn and van Dijk, 2005).

2.1 Change-point models with a known number of breaks

A conditionally linear change-point model with a known number of breaks is defined as

$$y_t = \beta'_m z_t + \sigma_m u_t$$

$$\beta_m = \beta_1, \ \sigma_m = \sigma_1, \text{ for } \tau_1 > t \ge 1$$

$$\dots$$

$$\beta_m = \beta_{M-1}, \ \sigma_m = \sigma_{M-1}, \text{ for } \tau_{M-1} > t \ge \tau_{M-2}$$

$$\beta_m = \beta_M, \sigma_m = \sigma_M, \text{ for } n \ge t \ge \tau_{M-1},$$
(3)

where $u_t \sim nid(0, 1)$, *M*-1 is the number of breaks and the unknown parameters are $(\beta_1, ..., \beta_M, \sigma_1, ..., \sigma_M, \tau_1, ..., \tau_{M-1})$. Chib (1998) obtains large efficiency gains over previous samplers by framing the change-point problem as

³A note on terminology is in order. Harvey (1989) uses the expression "conditionally Gaussian" to denote state-space models without indicators, but with system matrices which are functions of lags of the dependent variables. Shephard (1994), Carter and Kohn (1994) and Gerlach, Carter and Kohn (2000) call a model conditionally Gaussian if $p(y^{1,n}|K^{1,n})$ is Gaussian, which does not nest Harvey's definition. The proofs given in the appendix of this paper extend the results of Carter and Kohn (1994) and Gerlach, Carter and Kohn (2000), so we can define a model to be conditionally Gaussian if $p(y_t|y^{1,t-1}, K^{1,t})$ is Gaussian.

⁴It is possible to allow $E(u_t v_t) \neq 0$.

a switching-regime problem with transition probabilities constrained so that regimes come in a non-reversible sequence. This model can be written in the state-space form (1) and (2) by setting $K_t = m$, $g_t = \beta'(K_t)z_t$, $\gamma_t = \sigma(K_t)$, and modelling K_t as a discrete Markov process with restrictions on the transition probability matrix. Chib adapts the sampler of Carter and Kohn (1994) and draws K in O(nM) operations. He also shows how to compute the marginal likelihood so that the number of breaks can be treated as an unknown quantity in a model selection or model averaging context. Chib's method does not require the measurement equation to be linear or Gaussian, but it does require the probability of y_t to be readily available given parameters $\theta(K_t)$, and is therefore impractical for models with time-varying unobserved states such as measurement errors or seasonal components.

2.2 Change-point models with an unknown number of breaks

Pesaran, Pettenuzzo and Timmermann (2004) generalize the model of Chib (1998), but retain the assumption of a known number of breaks. Even though they integrate over the number of breaks M-1 by computing the marginal likelihood for $M = 0, 1, ..., M_{max}$, Koop and Potter (2004a) point out that there are several drawbacks to this strategy, besides the large computational costs. One is the difficulty in nesting the case of rare, large breaks, and of frequent, smaller breaks. A second problem is that the need to impose exactly M-1 in-sample breaks typically implies a prior in which breaks near the end of the sample are inordinately more likely than at earlier dates.

To avoid these shortcomings, Koop and Potter (2004b) develop a changepoint model with an unknown number of breaks. Letting the number of breaks be random is not only valuable per se, but also delivers a prior in which all observations have the same probability of being a break-point. Their key idea is to model an unknown number of breaks indirectly, by assuming that there are M regimes but that up to M-1 may occur out of sample. They set M = n and are therefore able to nest a time-varying-parameter model and a model with few breaks (given a sufficiently diffuse prior on regime duration).

Another innovation in Koop and Potter (2004b) is that regime duration has a Poisson prior in place of the standard geometric prior. A geometric prior for regime duration corresponds to the assumption of an independent breaking process, i.e. $p(K_t|K_{s\neq t}) = p(K_t)$, where $K_t = 1$ if there is a break at time tand $K_t = 0$ otherwise. The geometric distribution implies that, for any j > 1, durations of j are more likely than durations of j+1, a feature of the prior that is relaxed in Koop and Potter (2004b).⁵

Koop and Potter (2004b) draw the break dates conditional on parameters by adapting the algorithm of Chib (1998). Chib's algorithm, originally designed for a geometric prior on duration, requires O(nM) operations to draw all break points. Koop and Potter set M = n, resulting in $O(n^2)$ operations. Transition probabilities are constant for a geometric distribution, whereas they depend on the distance from the last break for a Poisson. The choice of a Poisson prior therefore requires averaging over durations in computing transition probabilities, leading to $O(n^3)$ operations, which limits the use of what is otherwise a flexible model. Section 4 shows how break points can be drawn in O(n) operations by adapting the sampling algorithm of Gerlach *et al.* (2000) rather than that of Chib (1998).

2.3 Time varying parameter models

Time-varying-parameter (TVP) models are written in the state-space form (1) and (2) with constant system matrices. Parameter variation comes through normal innovations in the transition equations. These models have a long tradition (West and Harrison (1997) is a classic Bayesian reference). Recent examples in economics include Cogley and Sargent (2001) and Primiceri (2005). Inference by Gibbs sampling is relatively straightforward, as a number of algorithms are available to draw the states conditional on parameters (including Carter and Kohn (1994), Fruhwirth-Schnatter (1994), de Jong and Shephard (1995) and Durbin and Koopman, 2002). The limitation of TVP models is that they are designed for smooth and frequent parameter variation and cannot effectively cope with infrequent interventions. Both types of breaks can be captured by mixture innovation models.

2.4 Mixture innovation models

Mixture innovation models are very general because they allow all system matrices in (1) and (2) to depend on K_t . Parameter shifts are typically modeled by letting $\Gamma_t \Gamma'_t$ (the covariance matrix of innovations to the states) depend on

⁵Koop and Potter (2004a) also show that prior regime duration remains approximately geometric if K_t is Markov. The idea of departing from a geometric distribution is not uncontroversial, though. Gelman *et al.* (1995, page 52) find that the geometric distribution's "'memoryless' property makes it a natural model for survival or lifetime data". Moreover, we note that the use of a Poisson prior may be questionable when breaks are thought to be (or turn out to be) relatively rare. For example, suppose that breaks occur every twenty years on average. On monthly data that corresponds to $\lambda = 240$, which implies a standard deviation of $\sqrt{240} \simeq 15.5$, or a little over a year. The Poisson prior therefore restricts infrequent breaks to recur with greater regularity than is probably prudent to assume.

 K_t . To illustrate, consider the simple example

$$y_t = \mu_t + \sigma_u u_t$$

$$\mu_t = \mu_{t-1} + K_t \sigma_v v_t,$$
(4)

where K_t is an independent sequence of Bernoulli variables, with

 $K_t = 1$ with probability π $K_t = 0$ with probability $1 - \pi$,

and u_t and v_t are both nid(0,1). This model can be written in terms of (1) and (2) by setting $g_t = 0$, $h_t = 1$, $\gamma_t = \sigma_u$, $f_t = 0$, $F_t = 1$, $\Gamma_t = \sigma_v K_t$. At each new observation, there is a probability π of a break; if there is a break, the change in the mean of the process is normally distributed with mean zero and standard deviation σ_v . The number of in-sample breaks is random, with prior distribution defined by π (possibly a random variable itself). Prior regime duration is geometric, with mean $(1-\pi)/\pi$ and standard deviation $\sqrt{(1-\pi)}/\pi$.⁶

This approach to structural change has a number of advantages over the change-point approach:

- 1. It is an intuitively simple way of allowing for random number and size of breaks, which facilitates prior elicitation.
- 2. Computational costs are linear in the number of values that K_t can take. This gives the modeler great flexibility. For example, we may wish to capture the idea that the mean of a process is unchanged in some periods, while in others it incurs small but relatively frequent and large but infrequent breaks. This can be achieved by setting

$$\sigma_v(K_t = K^i) = \sigma_{v_i}$$
, $prob(K_t = K^i) = \pi_i$ for $i = 1, 2, 3,$

with $\sigma_{v_1} = 0$, and a prior for which $\sigma_{v_3} \gg \sigma_{v_2}$ and $\pi_2 \gg \pi_3$. This is a general and intuitive way of nesting TVP models and models with infrequent breaks.⁷

3. It is just as easy to assume that parameters in different regimes (i) are drawn from a common distribution, as in Chib (1998) and Pesaran *et al.*

⁶Notice that a geometric distribution with mean μ has standard deviation close to μ , whereas a Poisson distribution with mean μ has standard deviation $\sqrt{\mu}$. When μ is relatively large, the two priors have very different implications for the regularity of occurrences of breaks.

⁷Koop and Potter (2004b) also nest a TVP model and a model with few breaks, given a sufficiently sparse prior on durations. However, mixture innovations are more general in that they allow to combine (rather than choose from) these two options.

(2004) (ii) depend only on parameters in the previous regime, as in Koop and Potter (2004b) and in most TVP and mixture innovation applications, or (iii) are a mixture of (i) and (ii). Referring to equation (4), the first case can be modelled as

$$\mu_t = mK_t + \mu_{t-1}(1 - K_t) + K_t \sigma_v v_t, \tag{5}$$

where K_t is either 0 or 1. To nest both cases, write

$$\mu_t = mK_{1t} + \mu_{t-1}(1 - K_{1t}) + (K_{1t}\sigma_{1v} + K_{2t}\sigma_{2v})v_t, \tag{6}$$

where $K_t = (K_{1t}, K_{2t})$, $K_t = (0, 0)$ for no break, and $K_t = (1, 0)$ and (0, 1) for a break with $\mu_t | K_t$ independent and a random walk respectively.

- 4. It is a convenient framework to allow different parameters to change at different times. Change-point models invariably assume that all parameters change at the same time (unless some are forced to be constant). Without this restriction, a change-point model would have M^p regimes to keep track of, where M-1 is the maximum number of breaks in each parameter and p is the number of parameters that are allowed to change, implying at least $O(nM^p)$ operations to draw the break dates. In contrast, if we let $K_{it} \in \{0, 1\}$ denote the absence or presence of a break in parameter i at time t, then $K_t = (K_{1t}, ..., K_{pt})$ can take 2^p values, so K can be drawn in $O(n2^p)$ operations regardless of the number of breaks (see Section 2.4.1).
- 5. It is a convenient framework to allow shifts in variance and in conditional mean parameters to occur independently (see Section 3).
- 6. It is a convenient framework to model breaks and outliers jointly. This can be important, as innovation and (especially) additive outliers may both hide actual breaks and spuriously indicate non-existing breaks, particularly in real-time forecasting.
- 7. While sampling schemes for multiple change-point models draw the break dates conditional on all parameters, in a mixture innovation approach most parameters can be treated as states and integrated out, which increases sampling efficiency. For example, referring to the model given by (3), we would write

$$y_t = \beta_t z_t + \sigma_t u_t$$

$$\beta_t = \beta_{t-1} + K_t \sigma_v v_t$$

treat β_t as the state vector and integrate it out when drawing K_t (see Section 2.4.1).

Finally, it is worth noting that mixture innovations can be combined with regime-switching and change-points (see Giordani *et al.* (2005) for an example), and that transition and break probabilities can be related to exogenous variables (see McCulloch and Tsay, 1993).

2.4.1 Sampling K in mixture innovation models

Mixture innovation models have been the subject of considerable study⁸, but until the mid-90s exact solutions were available only for the simplest cases and at great computational expense. Several approximations were proposed, including Harrison and Stevens (1976), Sims (1993) and Engle and Smith (1999). Markov chain Monte Carlo techniques opened the way for the exact solutions developed by McCulloch and Tsay (1993), Carter and Kohn (1994), and Shephard (1994). However, the algorithms proposed by Carter and Kohn (1994) and Shephard (1994) draw the auxiliary variables K conditional on the states x. In the case of structural breaks or additive outliers, K and x are highly correlated, making these samplers very inefficient. Often, as in the model given in (4), the correlation is perfect, and the samplers break down completely (see Gerlach et al., 2000). McCulloch and Tsay (1993) reduce the severity of this problem by drawing K_t conditional on the error v_t rather than on the state x_t , but their sampler remains rather inefficient (Gerlach et al., 2000). Gerlach et al. (2000) develop an algorithm for conditionally Gaussian processes that draws K without conditioning on x, and therefore retains a high degree of efficiency regardless of the correlation between K and x.

The first contribution of Gerlach *et al.* (2000) is to draw K from

$$p(K_t|y, K_{s\neq t}, \theta) \propto p(y^{t,n}|y^{1,t-1}, K, \theta) p(K_t|K_{s\neq t}, \theta), \tag{7}$$

where the states have been integrated out (rather than conditioned on) and θ is the vector of parameters. For a given proposed value of K_t , $p(K_t|K_{s\neq t},\theta)$ is evaluated from the transition probabilities, and $p(y^{t,n}|y^{1,t-1}, K, \theta)$ can be computed with the Kalman filter in conditionally Gaussian models. Evaluating $p(y^{t,n}|y^{1,t-1}, K, \theta)$ through the Kalman filter is straightforward, but requires O(n) operations, implying $O(n^2)$ operations to draw K. Gerlach *et al.* (2000)'s second contribution is to provide an algorithm to evaluate $p(y^{t,n}|y^{1,t-1}, K, \theta)$ in one step and thus to draw K in O(n) operations. Since K_t takes a finite number of values, it can be drawn by computing $p(K_t|y, K_{s\neq t}, \theta)$ for all possible values of K_t and then normalizing. The computing time is then linear in both n and

⁸Harrison and Stevens (1976) call them multiprocess models. See also Kitagawa (1987).

in the number of elements in the mixture (the number of values that K_t can take).

3 Breaks in conditional variance

A mixture innovation approach can also be used to model variance shifts, as first shown by McCulloch and Tsay (1993). Their algorithm draws K conditional on the states and is therefore inefficient. Drawing K without conditioning on the states as in Gerlach *et al.* (2000) is preferable. However, because squared residuals are not normal, it would seem that their algorithm is unsuitable for variance shifts. This problem is overcome as follows. We begin by considering shifts in variance in an otherwise white noise process, and then show how to combine shifts in variance with shifts in other parameters.

Modeling shifts in variance. Let y_t be a zero-mean, uncorrelated random variable such that

$$y_t = \sigma_t \epsilon_t, \tag{8}$$

where ϵ_t is standard normal. Then $log(y_t^2) = log(\sigma_t^2) + u_t$, where u_t is $log(\chi_1^2)$ distributed. Following the stochastic volatility literature, we work with $log(\sigma_t^2)$, ensuring that σ_t^2 is always positive. It is then natural to model permanent shifts in σ_t^2 as

$$log(\sigma_t^2) = log(\sigma_{t-1}^2) + \sigma_v(K_{2,t})v_t, \tag{9}$$

where v_t is a standard normal random variable. If K_{2t} takes only one value the model reduces to smoothly changing variances of the type used by Primiceri (forthcoming). In Section 6 we set $\sigma_v(K_{2,t} = 0) = 0$ and $\sigma_v(K_{2,t} = 1) = \sigma_v^* > 0$ with a prior favouring infrequent shifts. More elements can be added to the mixture, for example to allow for both smooth and sudden shifts in variance, analogously to the case of shifts in conditional mean parameters.

We follow Carter and Kohn (1993 and 1997), Shephard (1994) and Kim *et al.* (1998), who observe that the distribution of a $log(\chi_1^2)$ can be very accurately approximated by a mixture of normals with few components. (Carter and Kohn (1997) use five, Kim *et al.* (1998) seven.⁹) We can then write the model in conditionally Gaussian state-space form

$$z_{t} = g(K_{1,t}) + h_{t} + \gamma(K_{1,t})\eta_{t}$$
(10)
$$h_{t} = h_{t-1} + \sigma_{v}(K_{2,t})v_{t},$$

⁹Both papers report all parameters of the mixture. We use the parameters from Carter and Kohn (1997).

where $z_t = log(y_t^2)$, $h_t = log(\sigma_t^2)$, and η_t and v_t are standard normal. The mean and standard deviation of each component of the mixture approximating the distribution of u_t determine $g(K_{1,t})$ and $\gamma(K_{1,t})$ respectively, so these matrices contain no unknown parameters. The prior probabilities π_i of each component are also known. Thus

$$p(u_t) \simeq \sum_{i}^{I} \pi_i N(g_i, \gamma_i^2),$$

where $N(g_i, \gamma_i^2)$ is the normal density with mean g_i and variance γ_i^2 .

Shifts in conditional first and second moments. We use the following sampling scheme to deal with shifts in both the conditional mean and conditional variance parameters:

- 1. Initialize the sampler with a time series of conditional standard deviations $\sigma^{1,n}$ (for example $\sigma_1 = \sigma_2 = \ldots = \sigma_n$).
- 2. Given $\sigma^{1,n}$, the model is conditionally Gaussian. Draw interventions K_m as in Gerlach *et al.* (2000) or with the adaptive Metropolis-Hastings sampler described in Section 5. Conditional on K_m , draw the states x with any of the algorithms mentioned in Section 2.3. Conditional on y, K_m and x, compute residuals r_t . The definition of residuals is such that $std(r_t) = \sigma_t$.
- 3. Model interventions in $log(r_t^2)$ as explained above in this section. Referring to equation (10), draw $K_v = (K_1^{1,n}, K_2^{1,n})$. Given K_v , draw $log(\sigma_t^2)$ for t = 1, ..., n with the same algorithms used to draw x in step (2).
- 4. Go to (2).

In going from (2) to (3), the most convenient assumption is that that shifts in conditional variance (second block) are independent of shifts in conditional mean parameters (first block). It is possible to have a prior relating probabilities of interventions in the first and second block, but some care is required as an overly tight prior may induce a nearly reducible chain. Innovation and additive outliers are drawn in the first block and accounted for in computing the residuals used in the second block (see Section 6 for an example), therefore reducing the risk of isolated outliers being interpreted as variance shifts.

4 Non-Markov breaking processes

This sections shows how to adapt the algorithm of Gerlach *et al.* (2000) when K_t is not Markov, and how this allows drawing the break dates in O(n) operations

in the change-point model of Koop and Potter (2004b) with a Poisson prior on regime durations.

Although Gerlach *et al.* (2000) work with Markov interventions, their key results do not require this assumption. In particular, the algorithm to draw $p(y^{t,n}|K)$ in one step does not rely on K_t being Markov, as all operations are conditioned on K, and it is therefore suitable for any breaking process as long as it is possible to evaluate $p(K_t|K_{s\neq t},\theta)$ (or $p(K|\theta) \propto p(K_t|K_{s\neq t},\theta)$). The independence and Markov assumptions are simply the most convenient to perform this evaluation.

4.1 Poisson prior for regime duration

Following Koop and Potter (2004b), assume that regime duration follows a Poisson distribution. The duration of regime m is defined as $d_m = \tau_{m+1} - \tau_m$, where τ_{m+1} and τ_m are adjacent break dates. They then assume that $d_m - 1$ is Poisson distributed $Po(\lambda)$ with parameter λ .¹⁰ We now show how to draw Kin mixture innovation models with a Poisson prior for durations, and then how the same technique can be used to draw break dates in the change-point model of Koop and Potter (2004b). In both cases, only O(n) operations are required for conditionally Gaussian models.

4.1.1 Drawing K

Let $K_t = 1$ ($K_t = 0$) stand for the presence (absence) of a break at time t. We want to evaluate $p(K_t|K_{s\neq t})$ —where the dependence on parameters θ has been dropped for convenience—when durations are Poisson distributed with known parameter λ . Assuming that there is at least one break both before and after t, the probability of a break at t is given by

$$p(K_t = 1 | K_{s \neq t}) = p(d_{1t})p(d_{2t}),$$

where d_{1t} and d_{2t} are regime durations, defined as the number of periods between two adjacent breaks

$$d_{1t} = min(t - \tau_1; t > \tau_1, K_{\tau_1} = 1)$$

$$d_{2t} = min(\tau_2 - t; t < \tau_2, K_{\tau_2} = 1),$$

and the probability of no break at t is given by

$$p(K_t = 0 | K_{s \neq t}) = p(d_{3t}),$$

¹⁰Koop and Potter (2004b) use a hierarchial prior to allow the mean duration λ to depend on *m*. For our purposes, we can assume a single λ with no loss of generality.

where d_{3t} is the number of periods between the last break before period t and the first break after period t

$$d_{3t} = min(\tau_2 - \tau_1; \ \tau_2 > t > \tau_1, \ K_{\tau_1} = K_{\tau_2} = 1).$$

Finally, $p(d_{it})$, where i = 1, 2, 3, is evaluated as

$$p(d_{it}) \propto \frac{exp(-\lambda)\lambda^{d_{it}-1}}{(d_{it}-1)!}.$$

The only slight complication arises when there is no in-sample break either before or after t. A simple but unsatisfactory solution is to assume $K_0 = K_{n+1} = 1$. To illustrate a better solution, keep the assumption $K_0 = 1$ (simply for convenience of exposition). Initialize the algorithm with an arbitrary value of n^* , such as $n^* = n + 1$, where

$$n^* = min(\tau; \tau > n, K_{\tau} = 1).$$

Given n^* , sequentially update $K_1, ..., K_n$ as described above. n^* is then easily updated from the prior conditional on $n - \tau_l$, where τ_l is the last in-sample break date.

4.1.2 Application to Koop and Potter (2004b)

Referring to equation (7), having shown how to evaluate $p(K_t|y, K_{s\neq t}, \theta)$ with a Poisson prior on duration, we can use Gerlach *et al.* (2000) to compute $p(y^{t,n}|K, \theta)$ in one step if the model is conditionally Gaussian.¹¹ To illustrate, consider a simplified version of the benchmark model in Koop and Potter (2004b):

$$y_t = \beta_m z_t + \sigma_m u_t$$

$$\beta_m = \beta_{m-1} + \eta_t^{\beta}$$

$$ln(\sigma_m) = ln(\sigma_{m-1}) + \eta_t^{\sigma}$$

$$var(\epsilon_t) = 1, var(\eta_t^{\beta}) = \Omega_{\beta}, var(\eta_t^{\sigma}) = \omega_{\sigma},$$

where $d_m - 1 \sim Po(\lambda)$ and other priors can be omitted for our purposes. Koop and Potter draw all parameters conditional on the regimes and then the regimes conditional on the parameter sub-set $\theta = (\beta_1, ..., \beta_m, \sigma_1, ..., \sigma_m, \lambda)$. If we frame the problem as one of drawing break dates rather than regimes, then, conditional

¹¹Non-Gaussian change-point models include Chib (1998) and Koop and Potter (2004a).

on θ and on the break dates, the model can be written as (1)-(2) with all system matrices known, by setting

$$g_t=eta_m^{'}z_t,\ h_t=0,\ \gamma_t=\sigma_m,$$

This completes the description of how to draw K in O(n) operations using the sampler of Gerlach *et al.* (2000).

5 Adaptive Metropolis-Hastings sampling for discrete latent variable models

5.1 Discussion

This section describes a new adaptive sampler that speeds up the drawing of K. Since this is, to the best of our knowledge, the first application of adaptive sampling for discrete latent variables in a time series context, the main ideas are discussed in some detail. Because the discussion is in terms of drawing K in a conditionally Gaussian model, the methods developed in this section are applicable to mixture innovation models as well as regime switching and multiple change-point models with unobserved states and to the change-point model with unknown number of breaks of Koop and Potter (2004b). Moreover, they are ideally suited to detect outliers in both time series and cross sectional data.

The key idea of adaptive sampling is to use previous draws to form efficient proposal distributions for the application of the Metropolis-Hastings method. Precursors to full adaptive sampling limited this learning process to a subset of the burn-in period. This guarantees that the chain is Markov, and hence that the standard convergence properties of Metropolis-Hastings are unaffected by the initial learning period. A fully adaptive sampler, however, does not limit the learning to the burn-in period. Because the proposal distribution is no longer constant, but rather depends on the history of the draws, the resulting chain is not Markov, so the usual proof of ergodicity of the Metropolis-Hastings algorithm no longer applies. Establishing the properties of an adaptive MH sampler is a rather complex problem. Recent progress has been made by Gilks, Roberts and Sahu (1998), Haario, Saksamn and Tamminen (2001), Atchadé and Rosenthal (2003), and Nott and Kohn (forthcoming).

In particular, Nott and Kohn (*forthcoming*) establish conditions for the validity of an adaptive sampling scheme for discrete distributions. However, their argument is also valid for a compact space and so can be applied by suitably truncating the priors on the unknown parameters and states without affecting practical performance. Adaptation needs to be implemented with care to ensure ergodicity. Loosely speaking, the main requirements are that we use the entire history of draws rather than a moving window and that we constrain the proposal distribution so that no event in the sample space of K has zero probability.

The traditional use of partially (limited to the burn-in) adaptive schemes is to gain some knowledge of a non-standard distribution. Adaptive algorithms can of course be used to increase sampling efficiency in this case. However, the intuition of Nott and Kohn (*forthcoming*), which we extend to our timeseries framework, is that an adaptive Metropolis-Hastings algorithm can increase sampling efficiency even when direct sampling from the distribution of interest is possible. The key idea is to use information coming from the history of the draws to cheaply draw from an approximation rather than expensively draw from the exact distribution. The approximation is then used as the proposal distribution in a Metropolis-Hastings step.

The general problem of detecting regimes and interventions in time series models is ideally suited to incorporate these ideas, because (i) the distribution of $(K_t|Y, K_{s\neq t})$ is expensive to compute, particularly when the number of interventions is random (ii) in most periods, most of the probability mass is concentrated in a small sub-set of the sample space of K_t . For example, during the running of the chain, for many observations it soon becomes clear whether a break in that period is likely or unlikely, or if the observation belongs to, say, regime one or two. Sizable computational gains are therefore possible by incorporating this information into a proposal distribution for a Metropolis-Hastings algorithm, rather than drawing K_t from the exact but computationally expensive distribution.

5.2 Sampling algorithm

We now present our adaptive Metropolis-Hastings (MH) sampling algorithm to draw K efficiently. The algorithm is first stated and then discussed.

- 1. For m = 1, ..., d, generate $K^{(m)}$ (the m-th draw of K) using the algorithm of Gerlach *et al.* (2000). After drawing each $K^{(m)}$, update all the model parameters.
- 2. For m = d + 1, ..., M (where M is the total number of iterations), carry out (a)-(c) for t = 1, ..., n, and then (d).
 - (a) Using past draws $K^{(1)}, ..., K^{(m-2)}$, compute the percentage of $K_t = K^i$ for i = 1, ..., J, where J is the number of values that K_t can take.

Call each percentage $p_{i,t}$.

- (b) Define $\alpha_{it} = max(min(p_{it}, 1-\delta), \delta)$, for i = 1, ..., J, where $\delta > 0$ (e.g. 0.01 or 0.02). Then normalize α_{it} so that $\sum_{i=1}^{J} \alpha_{it} = 1$. Draw $P \in \{1, ..., J\}$ from a multinomial density with parameters $\{\alpha_{1t}, ..., \alpha_{Jt}\}$. Given P, form K^P .
- (c) Let C be the current indicator of K_t , that is, $K^C = K_t^{(m-1)}$. We accept $K_t^{(m)} = K^P$ with probability min(1, p), where

$$p = \frac{p(K_t = K^P | y, K_{s \neq t})}{p(K_t = K^C | y, K_{s \neq t})} \frac{\alpha_{Ct}}{\alpha_{Pt}}$$

and $p(K_t = K^P | y, K_{s \neq t})$ can be computed efficiently using the method of Gerlach *et al.* (2000), and (crucially) no computation is required whenever $K^P = K^C$.

(d) Update all model parameters.

We note that adaptation is carried out only on the latent variables K_t and not on the other parameters and variables in the model.

The learning mechanism outlined in step 2.a has two characteristics that ensure ergodicity: (i) an expanding (rather than fixed) window of draws (ii) a proposal distribution built with information up to m-2 only ($K^{(m-1)}$ is available but not used). The percentages mentioned in step 2.a may be updated at each iteration or only infrequently (we update them every 50 iterations). In step 2.b, a strictly positive constant δ ensures that we can draw with positive probability even those interventions that have not been drawn before (thus ensuring that the sampling scheme does not become reducible). Crucially, for a large proportion of the observations the multinomial density ($\alpha_{1t}, ..., \alpha_{Jt}$) typically assigns most of the probability mass to one or a few outcomes. For example, for most observations the probability of an outlier is either close to zero or to one. The probability of a break at t is rarely close to one, but is often very small. The proposal distribution built in 2.b is used in 2.c in a Metropolis-Hastings step.

5.3 Assessing efficiency gains

The efficiency gains of our adaptive MH sampler are likely to be larger when interventions are relatively rare (as for breaks) and/or relatively clear-cut (as for outliers and, in some cases, regime-switching), since then the large majority of proposals is of the type $K^P = K^C$, and hence no calculation is required in step 2.c. Computational gains are also likely to increase when J is large. To say more, we need to specify a model, a data-set and a prior, as the efficiency of our adaptive MH sampler can only be evaluated numerically. Since the number of objects of interest (including K) is high, we summarize sample draws through the distributions of selected parameters and of the states at t = 1. Denoting by θ_i^m the *m*-th draw of parameter *i*, we compare Algorithm 1 in Gerlach *et al.* (2000) with our adaptive MH sampler by (i) confirming that they lead to the same posterior inference, summarized by $\overline{\theta_i} = (\sum_{m=1}^M \theta_i^m)/M$ and $\widehat{\sigma_i}^2 = [\sum_{m=1}^M (\theta_i^m - \overline{\theta_i})^2]/M$, and (ii) comparing their relative efficiency. Par (ii) is done as follows: An estimate of the variance of $\overline{\theta_i}$ is

$$var(\overline{\theta_i}) = \frac{\widehat{\sigma_i}^2}{M} [1 + 2\sum_{i=1}^R (1 - \frac{j}{M})\widehat{\rho_i}(j)],$$

where $\hat{\rho_i}(j) = \widehat{corr}(\theta_i^m, \theta_i^{m-j})$ and the truncation point R is such that $\hat{\rho_i}(j) \simeq 0$ for j > R. Assuming that M is sufficiently large, $\hat{\sigma_i}^2$ is the same for both samplers. The term in square parenthesis is the inefficiency factor (IF). An independent sampler has unit inefficiency factor. We can couple the inefficiency factor with the running time of the two sampling schemes to obtain a measure of the relative efficiency of the adaptive scheme. The number of iterations needed for the adaptive scheme to have the same $var(\overline{\theta_i})$ as the non-adaptive scheme is given by the ratio of the inefficiency factors. We can then multiply this number by the ratio of computing time (CT) to obtain a measure of the relative efficiency of the adaptive Metropolis-Hastings sampler:

Relative efficiency =
$$\frac{\text{CT non-adaptive}}{\text{CT adaptive}} \frac{\text{IF non-adaptive}}{\text{IF adaptive}}$$
.

6 Efficiency gains in models of U.S. real interest rate and inflation

This section computes relative efficiencies in two applications to U.S. real interest rate and inflation. The models are meant to illustrate the mixture innovation approach to structural breaks and to evaluate the efficiency of the adaptive sampler rather than to provide a thorough empirical analysis of the two series.

In both cases 20000 iterations are used, with a burn-in of $100.^{12}$ The first 100 iterations are always performed with Algorithm 1 in Gerlach *et al.* (2000). After 100 iterations we either continue to use the same algorithm or switch to the adaptive Metropolis-Hastings sampler, with $\delta = 0.01$.

 $^{^{12}}$ The burn-in may seem low, but in our experience convergence to the stationary distribution occurs very quickly in this type of model using Algorithm 1 in Gerlach *et al.* (2000).

6.1 Mean shifts in the U.S. real interest rate

U.S. real interest rates have attracted considerable attention in the multiple change-point literature because they seem to display several sharp shifts in mean, at least in a univariate context (Bai and Perron, 1998). The data are quarterly from 1952Q1 to $2004Q4^{13}$. We consider an AR(1) model with fixed autoregressive coefficient and fixed residual variance, time variation modeled directly in the mean of the process rather than in the intercept, and additive outliers

$$y_{t} = y_{t}^{*} + \sigma_{e} K_{v,t} v_{t}$$

$$y_{t}^{*} = \mu_{t} + \rho(y_{t-1}^{*} - \mu_{t-1}) + \sigma_{e} e_{t}$$

$$\mu_{t} = \mu_{t-1} + \sigma_{e} K_{\mu,t} u_{t}$$

$$p(K_{t}) = p(K_{t} | K_{s \neq t}),$$
(11)

where $K_t = (K_{v,t}, K_{\mu,t})$. Additive outliers are modelled by letting $K_{v,t} \in \{0, g_v\}$ and breaks in mean by letting $K_{\mu,t} \in \{0, g_{\mu_1}, g_{\mu_2}\}$, where $(g_v, g_{\mu_1}, g_{\mu_2})$ are random. In total, K_t can take the four values (for given $g_v, g_{\mu_1}, g_{\mu_2}$), shown in Table 1.

The priors for $(g_v^2, g_{\mu_1}^2, g_{\mu_2}^2)$ are inverse gamma

$$g_i^2 \sim IG(S_i, n_i),$$

where $i \in \{v, \mu_1, \mu_2\}$, $n_i = 5$ and $\sqrt{S_v/n_v} = 2.5$, $\sqrt{S_{\mu_1}/n_{\mu_1}} = 1$, $\sqrt{S_{\mu 2}/n_{\mu_2}} = 3$. These priors are not very tight, but they do not try to be diffuse either. As shown in McCulloch (2000), being uninformative or nearly uninformative on the size of breaks ends up placing most prior probability on extremely large breaks, with the result that no break is ever found. Conversely, diffuse priors on the frequency of breaks place considerable probability on very frequent breaks, and therefore carry a large risk of over-fitting.

Letting π_i be the prior probability that $K_t = K^i$, the prior distribution for $(\pi_1, \pi_2, \pi_3, \pi_4)$ is Dirichelet

$$p(\pi_1, \pi_2, \pi_3, \pi_4) \sim D(n_0 \pi_1^*, \dots, n_0 \pi_4^*),$$

where n_0 can be interpreted as the number of prior observations (the tightness

¹³The nominal interest rate is the yield on three months government bonds, secondary market. Inflation is CPI inflation for urban consumers, all items, seasonally adjusted. Both series are from the database FRED II (http://research.stlouisfed.org/fred2), aggregated from monthly data.

of the prior), and $(\pi_1^*, ..., \pi_4^*)$ as their relative frequencies (the prior mean).¹⁴ To emphasize the strength of the evidence of mean shifts, we impose a tight prior of extremely infrequent breaks: $n_0 = 1000$, and the prior probabilities of $K_{\mu} = g_{\mu_1}$ and $K_{\mu} = g_{\mu_2}$ are both just 0.1%, implying an average regime duration of 125 years. In spite of such tight priors, posterior distributions show rather clear evidence of four mean shifts, corresponding closely with common historical interpretations of the data: low real interest rates during the inflation of the 70s, then a sudden increase with the Volcker deflation, and after a few years a return to lower values. Once mean shifts are taken into account, the real interest rate displays little persistence: Table 1 and Figure 1 show that the posterior mean of the autoregressive coefficient ρ is only 0.35.

Table 3 shows that the relative efficiencies of the MH sampler for μ_1 , ρ and σ_e are 3.13, 3.64, and 3.47 respectively. That is, on average, the adaptive sampling scheme reaches a given Monte Carlo standard error in about 29% of the time required by Algorithm 1 in Gerlach *et al.* (2000). The adaptive scheme is a bit less efficient per iteration, but the efficiency loss is compensated for by the increase in speed.

6.2 Changing mean, dynamics and volatility in U.S. inflation

Several studies (see, for example, Nelson and Piger, 2002) have suggested the presence of structural breaks in the mean of U.S. inflation.¹⁵ Moreover, there is some statistical evidence and some economic theory to support the nonconstancy of inflation persistence over long time periods (Akerlof *et al.* (2000), Christiano and Fitzgerald, 2003). However, other researchers have found little variation in either the intercept or autoregressive coefficients of AR models for U.S. inflation, but strong evidence of changes in residual variance (for example, Primiceri (*forthcoming*), and Koop and Potter, 2004b).

We model quarterly U.S. CPI^{16} inflation for the period 1951Q1-2004Q4 as an AR(1) process with random breaks in intercept, autoregressive parameter, and residual variance as:

¹⁴The posterior distribution of $g_v^2, g_{\mu_1}^2, g_{\mu_2}^2$ given σ_{ϵ} is inverse gamma. See Giordani *et al.* (2005) for details.

¹⁵This statement implicitly assumes a linear reference model. Sargent (1999) is an example of a non-linear model in which shifts in the mean and dynamics of inflation arise endogenously, a reminder that the presence or absence of structural breaks can only be evaluated in relation to a model.

¹⁶Consumer Price Index for all urban consumers (all items), from the database Fred II, series ID: CPIAUCSL, seasonally adjusted. CPI aggregated from monthly data (averages) and inflation defined as $400(\frac{CPI_t}{CPI_{t-1}}-1)$.

$$y_{t} = c_{t} + b_{t}y_{t-1} + \sigma_{t}K_{e,t}e_{t}$$

$$c_{t} = c_{t-1} + K_{c,t}u_{t}^{c}$$

$$b_{t} = b_{t-1} + K_{b,t}u_{t}^{b}$$

$$log(\sigma_{t}^{2}) = log(\sigma_{t-1}^{2}) + K_{v,t}v_{t}$$

$$p(K_{t}) \equiv p(K_{m,t}, K_{v,t}) = p(K_{m,t})p(K_{v,t}) = p(K_{t}|K_{s\neq t}),$$
(12)

where $K_{m,t} = (K_{e,t}, K_{c,t}, K_{b,t})$. The latent variable $K_{e,t}$ can take the values (1, 2.5), where 1 is a standard observation and 2.5 an innovation outlier; $K_{c,t}$ can take the values (0, 0.2, 1) and $K_{b,t}$ can take the values (0, 0.5). For ease of interpretation, we assume that a break and an outlier cannot occur simultaneously, but we do allow breaks in c_t and b_t to occur both separately and jointly. This last feature seems new to the literature. As discussed in Section 2.4, extending change-point models to relax the assumption that all parameters break at the same time is straightforward in principle, but problematic in practice, whereas in our case the increased computational burden is small. In total, the vector $K_{m,t}$ can take seven values (refer to Table 2). $K_{v,t}$ can take the values (0, 1.39); $\sigma_t = \sigma_{t-1}$ for $K_{v,t} = 0$, while $K_{v,t} = 1.39$ and $v_t = 1$ (-1) imply $\sigma_t/\sigma_{t-1} \simeq 2$ $(\sigma_t/\sigma_{t-1} \simeq 0.5)$.

The prior probabilities of interventions are fixed and reflect the assumption that breaks in any parameter are rare (the combined probability of a break in c_t and/or b_t is 1%, so the prior mean interval between breaks is 25 years; the probability of a break in variance is also 1%). Priors on all the other parameters are conditionally conjugate but diffuse, and centered on OLS values.

The results are summarized in Figure 2. The rise in inflation in the late sixties and seventies is captured by an increase in both coefficients, whereas the falling inflation of the eighties leaves the constant nearly unaffected but corresponds to a large and sudden decrease in persistence. Inflation persistence is a positive function of the inflation level, as observed by Akerlof *et al.* (2000). The conditional standard deviation shows marked changes during the sample, which are also strongly correlated with the inflation level.¹⁷

Table 3 shows that the relative efficiencies of the MH sampler for c_1, b_1 , and σ_1 are 1.94, 2.82, and 2.94 respectively.

 $^{^{17}}$ This finding is hardly surprising. There is conclusive evidence that inflation is less forecastable at higher levels (see Giordani and Söderlind (2003) for a recent discussion).

7 Conclusion

Koop and Potter (2004a and 2004b) convincingly illustrate the advantages of allowing for a random number of breaks in change-point models. Our article argues that a mixture innovation approach is often a more natural and general way of achieving the same goal, and that fast and reliable estimation of conditionally Gaussian mixture innovation models is possible using the algorithm of Gerlach *et al.* (2000). The sampler of Chib (1998) is efficient when the number of break-points is known, but can otherwise involve dramatically higher computational costs.

Our article also introduces to the time series literature the concept of adaptive Metropolis-Hastings sampling for discrete latent variable problems. A simple adaptive scheme produces large efficiency gains (arising from reduced computing times) in two empirical applications to models with outliers and shifts in both conditional mean and conditional variance parameters. When considering the generality and simplicity of the sampler and the small additional programming costs required, these efficiency gains suggest that adaptive Metropolis-Hastings for discrete latent variable models should prove a fruitful concept in time series.

Acknowledgments. The work of Paolo Giordani was mostly carried out at the University of New South Wales. The work of Robert Kohn was partially supported by an Australian Research Council Grant on mixture models.

References

- Akerlof, G., Dickens, W.T., and Perry, G.L., 2000, Near rational wage and price setting and the optimal rates of inflation and unemployment, *Brookings Papers on Economic Activity*, 1, 1-60.
- [2] Atchadé, Y., and Rosenthal, J., 2003. On adaptive Markov chain Monte Carlo algorithms, Technical Report, Department of Mathematics and Statistics, University of Montreal.
- [3] Bai, J., and Perron, P., 1998, Estimating and testing linear models with multiple structural changes, *Econometrica*, 66, 47-78.
- [4] Carter, C. K., and Kohn, R., 1993, On the applications of Markov chain Monte Carlo methods to linear state space models, in Proc. Bus. Econ. Statist. pp. 131-136.
- [5] Carter, C.K., and Kohn, R., 1994, On Gibbs sampling for state-space models, *Biometrika*, 83, 589-601.

- [6] Carter, C.K., and Kohn, R., 1997, Semiparametric Bayesian inference for time series with mixed spectra, *Journal of the Royal Statistical Society*, Series B, 255-68.
- [7] Chernoff, H., and Zacks, S. 1964, Estimating the current mean of a normal distribution which is subject to changes in time, *Annals of Mathematical Statistics*, 35, 999-1028.
- [8] Chib, S. 1998, Estimation and comparison of multiple change-point models, *Journal of Econometrics*, 86, 221-241.
- [9] Christiano, L.J, and Fitzgerald, T.J., 2003, Inflation and monetary policy in the twentieth century, *Chicago Fed Economic Perspectives*, 1, 1-24.
- [10] Clements, M.P., and Hendry, D.F., 1999, Forecasting non-stationary economic time series, Cambridge, Mass., MIT Press.
- [11] Cogley, T. and Sargent, T., 2001, Evolving post-World War II inflation dynamics, NBER Macroeconomic Annual.
- [12] de Jong, P. and Shephard, N., 1995, The simulation smoother for time series models, *Biometrika*, 82, 339-350.
- [13] Durbin, J., and Koopman, S., 2002, A simple and efficient simulation smoother for state space time series analysis, *Biometrika*, 89, 603-616.
- [14] Elliott, G., and Müller, U.K., 2003, Optimally testing general breaking processes in linear time series models, Working Paper, Princeton University.
- [15] Engle, R.F., and Smith A.D., 1999, Stochastic Permanent Breaks, *Review of Economics and Statistics*, 81, 553-574.
- [16] Fruhwirth-Schnatter, S., 1994, Data augmentation and dynamic linear models, *Journal of Time Series Analysis*, 15, 183-202.
- [17] Gelman, A., Carlin, J.B., Stern, H.S., and Rubin, D.B., 1995, Bayesian data analysis, Chapman and Hall, London.
- [18] Gerlach, R., C. Carter and R. Kohn, 2000, Efficient Bayesian inference for dynamic mixture models, *Journal of the American Statistical Association*, 95, 819–828.
- [19] Gilks, W.R., Roberts, G.O. and Sahu, S.K. 1998, Adaptive Markov chain Monte Carlo through regeneration, Journal of the American Statistical Association, 93, 1045-1054.

- [20] Giordani, P., and P. Söderlind, 2003, Inflation forecast uncertainty, European Economic Review, 47, 1037-59.
- [21] Giordani, P., Kohn, R., and van Dijk, D., 2005, A unified approach to nonlinearity, structural change and outliers, Manuscript, University of New South Wales.
- [22] Haario, H., Saksman, E. and Tamminen, J. 2001, An adaptive Metropolis algorithm, *Bernoulli*, 7, 223-242.
- [23] Hansen, B., 2001, The new econometrics of structural change: Dating breaks in U.S. labor productivity, *Journal of Economic Pespectives*, 15, 4, 117-128.
- [24] Harrison, J., and Stevens, C.F. 1976. Bayesian forecasting (with discussion), Journal of the Royal Statistical Association, B 38, 205-47.
- [25] Harvey, A.C., 1989, Forecasting, structural time series models and the Kalman filter, Cambridge University Press, Cambridge.
- [26] Kim, S., Shephard, N., and Chib, S., 1998, Stochastic volatility: Likelihood inference and comparison with ARCH models, *Review of Economic Studies*, 65, 361-393.
- [27] Kitagawa, G., 1987. Non-Gaussian state space modeling of non-stationary time series, *Journal of the American Statistical Association*, 82, 1032-41.
- [28] Koop, G., and Potter, S.M., 2004a, Prior elicitation in multiple changepoint models, Manuscript, University of Leicester.
- [29] Koop, G., and Potter, S.M., 2004b, Forecasting and estimating multiple change-point models with an unknown number of change-points, Manuscript, University of Leicester.
- [30] Levin, A.T., and Piger, J., 2003, Is inflation persistence intrinsic in industrial economies? Federal Reserve Bank of St. Louis Working Paper 2002-023E.
- [31] McCulloch, R.E., and Tsay, R.S., 1993, Bayesian inference and prediction for mean and variance shifts in autoregressive time series, *Journal of the American Statistical Association*, 88, 968-978.
- [32] McCulloch, J.H. 2000, State-space time series modeling of structural breaks, Working Paper number 00-11, Ohio State University.

- [33] Nott, D., and Kohn, R., *forthcoming*, Adaptive sampling for Bayesian variable selection, *Biometrika*.
- [34] Pesaran, H., Pettenuzzo, D., and Timmermann, A. 2004, Forecasting time series subject to multiple structural breaks, Manuscript, University of California, San Diego.
- [35] Primiceri, G., 2005, Time varying structural vector autoregressions and monetary policy, *Review of Economic Studies*, 72, 821-852.
- [36] Quandt, R., 1958, The estimation of the parameters of a linear regression system obeying two separate regimes, *Journal of the American Statistical Association*, 53, 873-880.
- [37] Sargent, T.J., 1999, The conquest of American inflation, Princeton, N.J., Princeton University Press
- [38] Shephard, N.E, 1994, Partial non-Gaussian state-space models, *Biometrika*, 81, 115-131.
- [39] Sims, C., 1993, A Nine-Variable Probabilistic Macroeconomic Forecasting Model, in Stock, J and Watson, M. (eds.) Business Cycles, Indicators and Forecasting University of Chicago Press, p. 179-204.
- [40] Stock, J., and Watson, M., 1996, Evidence on structural instability in macroeconomic time series relations, *Journal of Business and Economic Statistics*, 14, 11-30.
- [41] West, M., and Harrison, J., 1997, Bayesian forecasting and dynamic models, Springer-Verlag, New York.

A Carter and Kohn (1994) and Gerlach, Carter and Kohn (2000) when system matrices depend on lagged observations.

A.1 Gerlach, Carter and Kohn (2000).

We extend the proof in Gerlach, Carter and Kohn (2000), henceforth GCK, to show that their algorithm to draw the indicators $\{K_1, \ldots, K_n\}$ in O(n) operations can still be applied if the system matrices are functions of lags of the dependent variable. We also write all formulas assuming that y_t is a vector, rather than a scalar as in GCK.

For any variable z_t , let $z^{t,s} = \{z_t, \ldots, z_s\}$, and $z = \{z_1, \ldots, z_n\}$. Consider the state-space model

$$y_t = g_t + h'_t x_t + G_t u_t \tag{13}$$

$$x_t = f_t + F_t x_{t-1} + \Gamma_t v_t \tag{14}$$

where the system matrices $g_t, h_t, G_t, f_t, F_t, \Gamma_t$ are determined by $y^{0,t-1}, K_t$ and by a vector of unknown parameters, and where K_t is a random vector such that $p(K_t|K_{s\neq t})$ is known up to some parameters. (To simplify notation, dependence on parameters is omitted from all expressions.)¹⁸ Equations (13) and (14) imply the following:

$$p(y_t|y^{0,t-1}, x^{1,t}, K) = p(y_t|y^{0,t-1}, x_t, K_t)$$
(15)

$$p(x_t|y^{0,t-1}, x^{1,t-1}, K) = p(x_t|y^{0,t-1}, x_{t-1}, K_t)$$
(16)

The proof in GCK consists of four lemmas, which we now consider in turn, modifying each expression as required to allow the system matrices to also depend on $y^{0,t-1}$, rather than on K_t and parameters alone.

Lemma 1. Let $r_{t+1} \equiv \operatorname{var}(y_{t+1}|x_t, K^{1,t+1}, y^{0,t})$. Then the following hold:

$$E(y_{t+1}|y^{0,t}, x_t, K^{1,t+1}) = g_{t+1} + h'_{t+1}(f_{t+1} + F_{t+1}x_t)$$
(17)

$$r_{t+1} = h'_{t+1}\Gamma_{t+1}\Gamma'_{t+1}h_{t+1} + G_{t+1}G'_{t+1}$$
(18)

and

$$E(x_{t+1}|y^{0,t+1}, x_t, K) = a_{t+1} + A_{t+1}x_t + B_{t+1}y_{t+1}$$
(19)

$$var(x_{t+1}|y^{0,t+1},x_t,K) = C_{t+1}C'_{t+1}$$
(20)

 $^{^{18}}$ GCK allow correlation between the errors of the transition and observation equations, whereas (13)-(14) do not. The proof extends straightforwardly to the case of correlated errors.

where

$$a_{t+1} = (I - B_{t+1}h'_{t+1})f_{t+1} - B_{t+1}g_{t+1}$$
(21)

$$A_{t+1} = (I - B_{t+1}h'_{t+1})F_{t+1}$$
(22)

$$B_{t+1} = \Gamma_{t+1} \Gamma'_{t+1} h_{t+1} r_{t+1}^{-1}$$
(23)

$$C_{t+1}C'_{t+1} = \Gamma_{t+1}(I - \Gamma'_{t+1}h_{t+1}r_{t+1}^{-1}h'_{t+1}\Gamma_{t+1})\Gamma'_{t+1}.$$
(24)

The expression on the right hand side of (24) can be factored as $C_{t+1}C'_{t+1}$ where the matrix C_{t+1} is either null or has full column rank. Then we can write

$$x_{t+1} = a_{t+1} + A_{t+1}x_t + B_{t+1}y_{t+1} + C_{t+1}z_{t+1},$$

where $z_{t+1} \sim N(0, I)$, independent of x_t and y_{t+1} (conditional on K). **Proof.**

$$p(y_{t+1}|y^{0,t}, x_t, K) = \int p(y_{t+1}|x_{t+1}, y^{0,t}, K) p(x_{t+1}|x_t, y^{0,t}, K) dx_{t+1}$$
$$= \int p(y_{t+1}|x_{t+1}, y^{0,t}, K_{t+1}) p(x_{t+1}|x_t, y^{0,t}, K_{t+1}) dx_{t+1}.$$

Integration over x_{t+1} gives (17)-(18) since both densities in the integral are Gaussian. We also have

$$p(x_{t+1}|y^{0,t+1}, x_t, K) \propto p(y_{t+1}|y^{0,t}, x_{t+1}, x_t, K)p(x_{t+1}|y^{0,t}, x_t, K).$$

Simple algebra shows that this density is Gaussian in x_{t+1} and depends only on K_{t+1} with mean and variance given by (19) and (20).

Lemma 2. For t = 1, ..., n - 1, the density $p(y^{t+1,n}|x_t, K, y^{0,t})$ is independent of $K^{1,t}$ and can be expressed as

$$p(y^{t+1,n}|y^{0,t}, x_t, K) \propto \exp\{-\frac{1}{2}(x'_t\Omega_t x_t - 2\mu'_t x_t)\},\$$

where Ω_t and μ_t are computed recursively starting from

$$\Omega_n = 0, \quad \mu_n = 0,$$

and moving backward

$$\Omega_{t} = A'_{t+1}(\Omega_{t+1} - \Omega_{t+1}C_{t+1}D_{t+1}^{-1}C'_{t+1}\Omega_{t+1})A_{t+1} + F'_{t+1}h_{t+1}r_{t+1}^{-1}h'_{t+1}F_{t+1},$$
(25)
$$\mu_{t} = A'_{t+1}(I - \Omega_{t+1}C_{t+1}D_{t+1}^{-1}C'_{t+1})(\mu_{t+1} - \Omega_{t+1}(a_{t+1} + B_{t+1}y_{t+1}))$$
(26)
$$+ F'_{t+1}h_{t+1}r_{t+1}^{-1}(y_{t+1} - G_{t+1} - h'_{t+1}F_{t+1})$$

where

$$D_{t+1} = C'_{t+1}\Omega_{t+1}C_{t+1} + I.$$

Proof. The proof is by induction as in GCK. It is easy to check that the result is true for t = n - 1. Suppose it is also true for t = j + 2. Then,

$$p(y^{j+1,n}|x_j, y^{0,j}, K) = p(y^{j+2,n}|x_j, y^{0,j+1}, K)p(y_{j+1}|x_j, y^{0,j}, K).$$

From (13) and (14)

$$p(y_{j+1}|x_j, y^{0,j}, K) \propto |r_{t+1}|^{-1/2} \exp(-0.5w'_{t+1}r_{t+1}^{-1}w_{t+1}),$$

where

$$w_{t+1} = y_{t+1} - g_{t+1} - h'_{t+1}(f_{t+1} + F_{t+1}x_t)$$

and the first term on the right-hand-side is given by

$$p(y^{j+2,n}|x_j, y^{0,j+1}, K) = \int p(y^{j+2,n}|x_{j+1}, y^{0,j+1}, K) p(x_{j+1}|x_j, y^{0,j+1}, K) dx_{j+1}.$$

The first density in the integral depends only on $K^{j+1,n}$ and the second density depends only on K_{j+1} . Hence the rest of the proof follows GCK.

Lemma 3. Let $m_t \equiv E(x_t|y^{0,t}, K), V_t \equiv var(x_t|y^{0,t}, K)$ and $R_t \equiv var(y_t|y^{0,t-1}, K)$. The Kalman filter for the state-space model (13) and (14) is given by

$$R_t = h'_t F_t V_{t-1} F'_t h_t + h'_t \Gamma_t \Gamma'_t h_t + G_t G'_t$$

$$\tag{27}$$

$$m_t = (I - J_t h'_t)(f_t + F_t m_{t-1}) + J_t R_t^{-1}(y_t - g_t)$$
(28)

$$V_t = F_t V_{t-1} F'_t + \Gamma_t \Gamma'_t - J_t R_t^{-1} J'_t$$
(29)

where

$$J_t = F_t V_{t-1} F_t' h_t + \Gamma_t \Gamma_t' h_t.$$

The conditional density $p(y_t|y^{0,t-1}, K^{1,t})$ is given by

$$p(y_t|y^{0,t-1}, K^{1,t}) \propto |R_t|^{-1/2} \exp(-0.5v'_t R_t^{-1} v_t),$$
 (30)

where

$$v_t = y_t - g_t - h'_t f_t - h'_t F_t m_{t-1}.$$

We can write $V_t = T_t T'_t$, where the matrix T_t has full column rank (if $V_t \neq 0$) or is null (if $V_t = 0$). Then, conditional on K,

$$x_t = m_t + T_t \xi_t,$$

where $\xi_t \sim N(0, I)$ and independent of $y^{0,t}$.

Proof. By assumption, $p(x_t|y^{0,t-1}, K) = p(x_t|y^{0,t-1}, K^{1,t})$ and $p(y_t|y^{0,t-1}, K) = p(y_t|y^{0,t-1}, K^{1,t})$. Then,

$$p(x_t|y^{0,t}, K) = \frac{p(y_t|x_t, y^{0,t-1}, K_t)p(x_t|y^{0,t-1}, K^{1,t})}{p(y_t|y^{0,t-1}, K^{1,t})}$$

so that

$$p(x_t|y^{0,t}, K) = p(x_t|y^{0,t}, K^{1,t}),$$

which is a normal distribution with mean m_t and variance V_t as given by (28) and (29). In addition,

$$p(x_{t+1}|y^{0,t},K) = \int p(x_{t+1}|x_t, y^{0,t}, K_{t+1}) p(x_t|y^{0,t}, K^{1,t}) dx_t$$

so that $p(x_{t+1}|y^{0,t}, K) = p(x_{t+1}|y^{0,t}, K^{1,t+1})$. Finally,

$$p(y_t|y^{0,t-1},K) = \int p(y_t|x_t, y^{0,t-1}, K_t) p(x_t|y^{0,t-1}, K^{1,t}) dx_t$$

only depends on $K^{1,t}$ and is computed as in Lemma 1.

Lemma 4.

$$p(y^{t+1,n}|y^{1,t},K) = \int p(y^{t+1,n}|y^{0,t},x_t,K^{t+1,n})p(\xi_t|K^{1,t})d\xi_t$$

$$\propto |T_t'\Omega_t T_t|^{-1/2} \exp\{-\frac{1}{2}(m_t'\Omega_t m_t - 2\mu_t'm_t - \phi_t'(T_t'\Omega_t T_t + I)^{-1}\phi_t)\},$$
(31)

where

$$\phi_t = T_t'(\mu_t - \Omega_t m_t).$$

Proof. Using Lemmas 2 and 3,

$$-2\log\{p(y^{t+1,n}|x_t, y^{1,t}, K^{t+1,n})p(\xi_t|K^{1,t})\} = x_t'\Omega_t x_t - 2\mu_t' x_t + \xi_t'\xi_t = (m_t + T_t\xi_t)'\Omega_t(m_t + T_t\xi_t) - 2\mu_t'(m_t + T_t\xi_t) + \xi_t'\xi_t.$$

Equation (31) is obtained by completing the squares and integrating out ξ_t .

A.2 Carter and Kohn (1994) when system matrices and transition probabilities depend on lagged obvervations.

We extend the proof in in Carter and Kohn (1994) to show that their algorithms to draw the states $\{x_1, \ldots, x_n\}$ conditional on the indicators $\{K_1, \ldots, K_n\}$ (and viceversa) in one block can still be applied if latent values of the dependent variables affect one or both of (i) the system matrices (ii) the probability distribution of the indicators. To draw the indicators in one block conditional on the states we need to assume that K_t is Markov conditional on $y^{0,t-1}$, that is $p(K_t|K^{1,t-1}, y^{0,t-1}) = p(K_t|K_{t-1}, y^{0,t-1}).$

Consider the state-space model given in (13) and (14), where the system matrices $g_t, h_t, f_t, G_t, F_t, \Gamma_t$ are determined by $y^{0,t-1}$, K_t and a vector of unknown parameters, and where K_t is a random vector such that $p(K_t|y^{0,t-1}, K_{s\neq t})$ is known up to some parameters.

Lemma 5: Generating the state vector. The following holds:

$$p(x|y) = p(x_n|y, K) \prod_{t=1}^{n} p(x_t|y^{0,t}, x_{t+1}, K_t),$$

where all densities are Gaussian. We can then generate x_n from a Gaussian with mean and variance given by the Kalman filter. To generate x_t given $x^{t+1,n}$, we observe that $x_t|y^{0,t}, K^{1,t}$ and $x_{t+1}|y^{0,t}, K^{1,t+1}$ are jointly normal

$$\begin{bmatrix} x_{t+1}|y^{0,t}, K^{1,t+1} \\ x_t|y^{0,t}, K^{1,t+1} \end{bmatrix} \sim N\left(\begin{bmatrix} f_{t+1} + F_{t+1}m_t \\ m_t \end{bmatrix}, \begin{bmatrix} F_{t+1}V_tF'_{t+1} + \Gamma_{t+1}\Gamma'_{t+1} & F_{t+1}V_t \\ V_tF'_{t+1} & V_t \end{bmatrix} \right)$$

(where m_t and V_t are defined in Lemma 3). Hence we can draw x_t from $p(x_t|x_{t+1}, y^{0,t}, K^{1,t+1})$, which is normal with known moments.

Proof. Assume that $p(x_1|y_0, x_0, K_1)$ is Gaussian. Then $p(x_{t+1}|y^{0,t}K^{1,t+1})$ is also Gaussian, since

$$p(x_{t+1}|y^{1,t}, K^{1,t+1}) = \int p(x_{t+1}|x_t, y^{1,t}, K_{t+1}) p(x_t|y^{1,t}, K^{1,t+1}) dx_t$$
$$= \int p(x_{t+1}|x_t, y^{1,t}, K_{t+1}) p(x_t|y^{1,t}, K^{1,t}) dx_t$$

because

$$p(x_t|y^{1,t}, K^{1,t+1}) \propto p(K_{t+1}|K^{1,t}, y^{1,t}) p(x_t|y^{1,t}, K^{1,t})$$

= $p(K_{t+1}|K_t, y^{1,t}) p(x_t|y^{1,t}, K^{1,t}) \propto p(x_t|y^{1,t}, K^{1,t})$.

Moreover, since

$$p(x_{t+1}|y^{1,t+1}, K^{1,t+1}) \propto p(y_{t+1}|x_{t+1}, y^{1,t}, K_{t+1})p(x_{t+1}|y^{1,t}, K^{1,t+1}),$$

the Kalman filter equations for x_t given in Lemma 3 hold. It is also easy to show that

 $p(x_t|y^{0,n}, x^{t+1,n}, K^{1,n}) = p(x_t|x_{t+1}, y^{0,t}, K^{1,t+1}).$

Lemma 5 follows straightforwardly. \blacksquare

Lemma 6: Generating the indicator variables. Assume that K_t is Markov conditional on $y^{0,t-1}$

$$p(K_t|y^{0,t-1}, x^{1,t-1}, K^{1,t-1}) = p(K_t|K_{t-1}, y^{0,t-1}).$$

Then we have

$$p(K|y) = p(K_n|y, x) \prod_{t=1}^n p(K_t|y^{1,t}, K_{t+1}, x_t).$$

Therefore we can generate K as in Carter and Kohn (1994), where the filtering equations work on $p(K_t|x_t, y^{1,t})$ rather than on $p(K_t|x_t, y_t)$. **Proof.** Suppose that we have evaluated $p(K_1|y^{0,1}, x_1)$. Then,

$$p(K_{t+1}|y^{0,t+1}, x^{1,t+1}) \propto p(y_{t+1}|x_{t+1}, y^{0,t}, K_{t+1})p(x_{t+1}|x_t, y^{1,t}, K_{t+1})p(K_{t+1}|y^{0,t}, x^{1,t})$$
$$p(K_{t+1}|y^{0,t}, x^{1,t}) = \sum_{K_t} p(K_{t+1}|K_t, y^{0,t})p(K_t|y^{0,t}, x^{1,t}).$$

We also have

$$p(K_t|K^{t+1,n}, x^{1,n}, y^{0,n}) = p(K_t|K_{t+1}, x^{1,t+1}, y^{0,t+1})$$

$$\propto p(y_{t+1}|x_{t+1}, y^{0,t}, K_{t+1})p(x_{t+1}|x_t, y^{0,t}, K_{t+1})p(K_t|y^{0,t}, x^{1,t}, K_{t+1})$$

$$\propto p(K_t|y^{0,t}, x^{1,t}, K_{t+1}).$$

Therefore we can generate ${\cal K}_t$ from

$$p(K_t|K^{t+1,n}, x^{1,n}, y^{0,n}) \propto p(K_{t+1}|K_t, y^{0,t})p(K_t|y^{0,t}, x^{1,t}).$$

	K^1	K^2	K^3	K^4
$K_{v,t}$	0	g_v	0	0
$K_{\mu,t}$	0	0	$g_{\mu 1}$	$g_{\mu 2}$
prior prob	0.978	0.020	0.001	0.001
post. prob	0.975	0.019	0.002	0.004
	$E(\cdot y)$	$std(\cdot y)$		
g_v	2.48	0.82		
g_{μ_1}	1.19	0.53		
$g_{\mu 2}$	3.57	1.52		
ρ	0.35	0.08		

Table 1: Values of K(t) and probabilities for model of the U.S. real interest rate. Values of K(t) in columns. First row: value of Kv (additive outlier). Second row: value of Km (break in mean). Third row: prior probability of each value of K(t). Fourth row: posterior probability of each value of K(t). Further rows: mean and std of model parameters.

	K^1	K^2	K^3	K^4	K^5	K^6	K^7
$K_{e,t}$	1	2.5	1	1	1	1	1
$K_{c,t}$	0	0	0.2	0.2	1	0	1
$K_{b,t}$	0	0	0	0.2	0	0.2	0.2
prob	0.97	0.02	0.002	0.002	0.002	0.002	0.002

Table 2: Values of Km(t) and probabilities for AR(1) model of U.S. inflation. Values of K(t) in columns. First row: value of Ke (innovation outlier). Second row: value of Kc (break in constant). Third row: value of Kb (break in autoregressive parameter). Kv(t) can take values 0 and 1.39 with prior probabilities 0.99 and 0.01, independent of Ke, Kc, Kb.

Inflation	IF	IF ratio	RE	Real rate	IF	IF ratio	RE
c_1	3.32	1.76	1.94	μ_1	1.26	1.14	3.13
b_1	2.14	1.21	2.82	ρ	1.68	0.98	3.64
σ_1	2.89	1.16	2.94	σ	1.58	1.03	3.47

Table 3: **Inefficiency factors.** Inefficiency factors of the adaptive MH algorithm, the ratio of the inefficiency factors of the adaptive MH algorithm over the algorithm of Gerlach et al. (2000), and the relative efficiency for selected parameters in two models of U.S. inflation and real interest rate.



Figure 1: AR(1) model with mean shifts for the U.S. real interest rate. (a) real interest rate and posterior mean of μ_t (b) posterior density of ρ (c) posterior mean of $K_{m,t}$ (d) posterior distribution of μ_{1976Q1} (e) posterior distribution of $\mu_{19822Q1}$ (f) posterior distribution of μ_{2004Q4} .



Figure 2: AR(1) model of U.S. inflation: (a) posterior mean of c_t (b) posterior mean of b_t (c) inflation and posterior median of $c_t/(1-b_t)$ (d) posterior mean of σ_t (e) posterior distribution of b_{1979Q1} (f) posterior distribution of b_{2004Q4} .

Earlier Working Papers:

For a complete list of Working Papers published by Sveriges Riksbank, see www.riksbank.se

Evaluating Implied RNDs by some New Confidence Interval Estimation Techniques by <i>Magnus Andersson</i> and <i>Magnus Lomakka</i>	. 2003:146
Taylor Rules and the Predictability of Interest Rates by <i>Paul Söderlind</i> , <i>Ulf Söderström</i> and <i>Anders Vredin</i>	2003:147
Inflation, Markups and Monetary Policy by Magnus Jonsson and Stefan Palmqvist	. 2003:148
Financial Cycles and Bankruptcies in the Nordic Countries by Jan Hansen	2003:149
Bayes Estimators of the Cointegration Space by Mattias Villani	2003:150
Business Survey Data: Do They Help in Forecasting the Macro Economy? by Jesper Hansson, Per Jansson and Mårten Löf	. 2003:151
The Equilibrium Rate of Unemployment and the Real Exchange Rate: An Unobserved Components System Approach by <i>Hans Lindblad</i> and <i>Peter Sellin</i>	2003:152
Monetary Policy Shocks and Business Cycle Fluctuations in a Small Open Economy: Sweden 1986-2002 by Jesper Lindé	2003:153
Bank Lending Policy, Credit Scoring and the Survival of Loans by Kasper Roszbach	2003:154
Internal Ratings Systems, Implied Credit Risk and the Consistency of Banks' Risk Classification Policies by <i>Tor Jacobson, Jesper Lindé</i> and <i>Kasper Roszbach</i>	2003:155
Monetary Policy Analysis in a Small Open Economy using Bayesian Cointegrated Structural VARs by <i>Mattias Villani</i> and <i>Anders Warne</i>	2003:156
Indicator Accuracy and Monetary Policy: Is Ignorance Bliss? by Kristoffer P. Nimark	2003:157
Intersectoral Wage Linkages in Sweden by Kent Friberg	2003:158
Do Higher Wages Cause Inflation? by Magnus Jonsson and Stefan Palmqvist	2004:159
Why Are Long Rates Sensitive to Monetary Policy by Tore Ellingsen and Ulf Söderström	2004:160
The Effects of Permanent Technology Shocks on Labor Productivity and Hours in the RBC model by <i>Jesper Lindé</i>	2004:161
Credit Risk versus Capital Requirements under Basel II: Are SME Loans and Retail Credit Really Different? by <i>Tor Jacobson, Jesper Lindé</i> and <i>Kasper Roszbach</i>	2004:162
Exchange Rate Puzzles: A Tale of Switching Attractors by <i>Paul De Grauwe</i> and <i>Marianna Grimaldi</i>	2004:163
Bubbles and Crashes in a Behavioural Finance Model by <i>Paul De Grauwe</i> and <i>Marianna Grimaldi</i>	2004:164
Multiple-Bank Lending: Diversification and Free-Riding in Monitoring by <i>Elena Carletti, Vittoria Cerasi</i> and <i>Sonja Daltung</i>	2004:165
Populism by Lars Frisell	2004:166
Monetary Policy in an Estimated Open-Economy Model with Imperfect Pass-Through by Jesper Lindé, Marianne Nessén and Ulf Söderström	2004:167
Is Firm Interdependence within Industries Important for Portfolio Credit Risk? by Kenneth Carling, Lars Rönnegård and Kasper Roszbach	2004:168
How Useful are Simple Rules for Monetary Policy? The Swedish Experience by Claes Berg, Per Jansson and Anders Vredin	2004:169
The Welfare Cost of Imperfect Competition and Distortionary Taxation by <i>Magnus Jonsson</i>	2004:170
A Bayesian Approach to Modelling Graphical Vector Autoregressions by Jukka Corander and Mattias Villani	2004:171
Do Prices Reflect Costs? A study of the price- and cost structure of retail payment services in the Swedish banking sector 2002 by <i>Gabriela Guibourg</i> and <i>Björn Segendorf</i>	2004:172
Excess Sensitivity and Volatility of Long Interest Rates: The Role of Limited Information in Bond Markets by <i>Meredith Beechey</i>	2004:173
State Dependent Pricing and Exchange Rate Pass-Through by Martin Flodén and Fredrik Wilander	2004:174
The Multivariate Split Normal Distribution and Asymmetric Principal Components Analysis by <i>Mattias Villani</i> and <i>Rolf Larsson</i>	2004:175
Firm-Specific Capital, Nominal Rigidities and the Business Cycle by David Altig, Lawrence Christiano, Martin Eichenbaum and Jesper Lindé	2004:176
Estimation of an Adaptive Stock Market Model with Heterogeneous Agents by <i>Henrik Amilon</i> Some Further Evidence on Interest-Rate Smoothing: The Role of Measurement	2005:177
Errors in the Output Gap by Mikael Apel and Per Jansson	2005:178

Bayesian Estimation of an Open Economy DSGE Model with Incomplete Pass-Through by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	. 2005:179
Are Constant Interest Rate Forecasts Modest Interventions? Evidence from an Estimated Open Economy DSGE Model of the Euro Area by <i>Malin Adolfson</i> , <i>Stefan Laséen, Jesper Lindé</i> and <i>Mattias Villani</i>	.2005:180
Inference in Vector Autoregressive Models with an Informative Prior on the Steady State by <i>Mattias Villani</i>	. 2005:181
Bank Mergers, Competition and Liquidity by <i>Elena Carletti, Philipp Hartmann</i> and <i>Giancarlo Spagnolo</i>	. 2005:182
Testing Near-Rationality using Detailed Survey Data by <i>Michael F. Bryan</i> and <i>Stefan Palmqvist</i>	. 2005:183
Exploring Interactions between Real Activity and the Financial Stance by <i>Tor Jacobson, Jesper Lindé</i> and <i>Kasper Roszbach</i>	.2005:184
Two-Sided Network Effects, Bank Interchange Fees, and the Allocation of Fixed Costs by <i>Mats A. Bergman</i>	. 2005:185
Trade Deficits in the Baltic States: How Long Will the Party Last? by <i>Rudolfs Bems</i> and <i>Kristian Jönsson</i>	. 2005:186
Real Exchange Rate and Consumption Fluctuations follwing Trade Liberalization by <i>Kristian Jönsson</i>	. 2005:187
Modern Forecasting Models in Action: Improving Macroeconomic Analyses at Central Banks by Malin Adolfson, Michael K. Andersson, Jesper Lindé, Mattias Villani and Anders Vredin	.2005:188
Bayesian Inference of General Linear Restrictions on the Cointegration Space by Mattias Villani	.2005:189
Forecasting Performance of an Open Economy Dynamic Stochastic General Equilibrium Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	.2005:190
Forecast Combination and Model Averaging using Predictive Measures by Jana Eklund and Sune Karlsson	. 2005:191
Swedish Intervention and the Krona Float, 1993-2002 by Owen F. Humpage and Javiera Ragnartz	. 2006:192
A Simultaneous Model of the Swedish Krona, the US Dollar and the Euro by Hans Lindblad and Peter Sellin	. 2006:193
Testing Theories of Job Creation: Does Supply Create Its Own Demand? by Mikael Carlsson, Stefan Eriksson and Nils Gottfries	. 2006:194
Down or Out: Assessing The Welfare Costs of Household Investment Mistakes by Laurent E. Calvet, John Y. Campbell and Paolo Sodini	. 2006:195



Sveriges Riksbank Visiting address: Brunkebergs torg 11 Mail address: se-103 37 Stockholm

Website: www.riksbank.se Telephone: +46 8 787 00 00, Fax: +46 8 21 05 31 E-mail: registratorn@riksbank.se